Author's response to reviews

Title: Spontaneous hyphaema secondary to bleeding from an iris vascular tuft in a patient with a supratherapeutic International Normalised Ratio: case report

Authors:

Kenneth G-J. Ooi (kgjooi@yahoo.com.au)
Rohan Gupta (rohangupta19@gmail.com)
Sarah B. Wang (swan7117@uni.sydney.edu.au)
Samuel P.G. Dance (spgdance@gmail.com)
Armand Borovik (aborovik@gmail.com)
Ian C. Francis (iancfrancis@gmail.com)

Version: 5 Date: 2 June 2015

Author's response to reviews:

Editor's Additional Request:
1. Please remove consent statement from the title page - DONE
2. Please move ethics statement for the title page to the methods section - DONE
3. Requesting for CARE Checklist - See Attached